Author's response to reviews

Title: The impact of supported telehealthcare monitoring in people with type 2 diabetes: study protocol for a randomized controlled trial

Authors:

Sarah Wild (sarah.wild@ed.ac.uk)
Janet Hanley (j.hanley@napier.ac.uk)
Steff Lewis (steff.lewis@ed.ac.uk)
John McKnight (john.mcknight@nhs.net)
Lucy McCloughan (lucy.mccloughan@ed.ac.uk)
Paul Padfield (paul.padfield@scotland.gsi.gov.uk)
Mary Paterson (mary.paterson@ed.ac.uk)
Hilary Pinnock (hilary.pinnock@ed.ac.uk)
Brian McKinstry (brian.mckinstry@ed.ac.uk)

Version: 3 Date: 6 June 2013

Author's response to reviews:

Dear Dr Sourij,

Re: MS: 1917263875952158 - The impact of supported telehealthcare monitoring in people with type 2 diabetes: study protocol for a randomized controlled trial

Thank you for giving us the opportunity to respond to your request to add a discussion section to our protocol. We agree that this would be a useful addition and have added a paragraph at the end of the paper.

We hope the revised paper will meet with your approval and look forward to hearing from you.

Yours sincerely,

Brian McKinstry (for Telescot group)